Author's response to reviews

Title: Combined inhibition of the cell cycle related proteins Wee1 and Chk1/2 induces synergistic anti-cancer effect in melanoma

Authors:

Gry I Magnussen (gry.magnussen@rr-research.no)
Elisabeth Emilsen (eee@ous-hf.no)
Karianne G Fleten (Karianne.Giller.Fleten@rr-research.no)
Birgit Engesæter (Birgit.Engesater@rr-research.no)
Viola Nähse-Kumpf (Viola.Naehse@rr-research.no)
Roar Fjær (Roar.Fjar@medisin.uio.no)
Ana Slipicevic (Ana.Slipicevic@rr-research.no)
Vivi A Flørenes (vivi.ann.florenes@radiumhospitalet.no)

Version: 4 Date: 11 May 2015

Author's response to reviews:

Dear Editor. Thank you for giving us the opportunity to publish our paper in BMC Cancer.

In response to editors request, a competing interests section as well as an authors' contribution sections have been included on page 18. Additionally, files have been removed